

### ASSET PRICING, ASSET ALLOCATION AND RISK-ADJUSTED PERFORMANCE WITH MULTIPLE GOALS AND AGENCY: THE GOALS AND RISK-BASED ASSET PRICING MODEL

Arun Muralidhar©

Investment managers require a consistent asset pricing model, asset allocation recommendations, and risk-adjusted performance measures (or the "three facets of investing") to be effective in managing portfolios. Incorporating three critical realities of investing into these models (i.e., that investors have many stochastic goals, seek to delegate to skillful agents, and maximize risk-adjusted returns) provides recommendations on the three facets that are different from the foundational papers of Modern Portfolio Theory (MPT). This paper briefly surveys the literature on MPT, Goals-based Investing (GBI), and agency before providing a normative Goals- and Risk-Based Asset Pricing Model (GRAPM) that includes these three realities of investing and articulates the three facets. GRAPM exploits a simple idea that a relatively risk-free asset for one stochastic goal is a risky asset for another, and vice versa. These two assets, plus the traditional absolute risk-free rate of MPT, allow us to triangulate to establish returns for all other assets based on the return of any goal-replicating asset and multiple correlations. This approach creates a "pair-wise equilibrium" for all assets (and potentially a general equilibrium)—different from MPT and also lends itself easily to a new asset pricing model with heterogeneous investors (i.e., each investor has a unique goal). GRAPM incorporates a "risk aversion" parameter that is also easily observable, and appears to explain why seemingly similar investors can have markedly different asset allocations or expected returns.



"When it is obvious that the goals cannot be reached, don't adjust the goals, adjust the action steps." Confucius<sup>1</sup>

#### 1 Introduction: Three Key Facets of Good Theory and Importance of Goals and Agency

Investment managers require a robust and consistent method to price, risk-adjust, and allocate assets, as these are the three key facets of an effective investment practice. Thus, effective theory should provide similarly consistent models and recommendations on asset pricing, risk-adjusted performance measures, and asset allocation. The foundational papers of Modern Portfolio Theory (MPT) provide these three facets, though in a piecemeal fashion. Markowitz (1952) and Tobin (1958) provide for asset allocation recommendations, Sharpe (1964), Lintner (1965), Mossin (1966), and Treynor (1966) provide the asset pricing model, and Treynor (1966), Sharpe (1994), Graham and Harvey (1994, 1997), and Modigliani and Modigliani (1997) provide riskadjusted performance measures for MPT. While each of these papers tackled these three key facets piecemeal/individually, Perold (2004), following Lintner (1965), demonstrates how MPT integrates all three facets and has a unique duality (under certain assumptions); namely, that the Capital Asset Pricing Model (CAPM) and Two Fund Separation (2FS) can be derived from maximizing the Sharpe Ratio or alternatively, that the maximization of the expected utility of wealth leads to CAPM, 2FS, and the Sharpe Ratio. This duality and completeness of approach on all three facets has yet to be matched by any other theory and therefore sets a very high bar to replace these foundational models.

MPT/CAPM is a normative theory that assumes that investors focus on two key variables:

expected returns and risk (defined as the absolute volatility of returns). Since MPT CAPM formulas are widely understood and adopted they are not repeated, but the model assumes that a riskfree asset, F, exists (i.e., one with zero volatility of returns and zero correlation to other assets) in net zero supply, with an exogenously determined expected return. It also assumes that a "market portfolio", M (i.e., a market-capitalization weighted portfolio of all risky assets), can be established and identified. Thereafter, all risky assets can be priced based on F, M and beta, but interestingly, the risky asset being priced is also a constituent of M. The theoretical pros and cons of MPT/CAPM are covered extensively in Levy (2011), but the fact that it is still used in practice globally bear testimony to the enduring power of this theory. However, MPT assumes a world distanced from reality to achieve this impressive result.

Subsequent theory recognizes that MPT overlooks two key realities of investing: first, that investors have multiple stochastic goals for which they are saving money (called Goals-based Investing or GBI as in Amenc et al., 2010; Das et al., 2010), and furthermore, that principals delegate investment decisions to agents (Brennan, 1993). Both nuances of investment practice meaningfully alter the asset pricing and asset allocation models, (and, in turn, the risk-adjusted performance measures to be used). Yet, many investors like pension funds globally, with a demonstrable stochastic pension benefit and multiple levels of delegation, have continued to use the traditional models of MPT and have failed to achieve their goals. Cumbo and Wigglesworth (2019) document how pension funds in the US and Netherlands are struggling in the low interest rate environment. In the last decade, their financial condition has declined dramatically: from having more than adequate assets to pay pensions in the late 1990s, to having to potentially institute pension cuts or take excessive risk (by investing in illiquid assets) to meet their goals.

Merton (2007) predicted these problems arguing that pension funds were mispricing the risk of their portfolios; additionally, the asset allocation recommendations of Sharpe and Tint (1990) were largely ignored. More specifically, Muralidhar (2019a, 2019b) argues that these pension crises have been caused because investors incorrectly used MPT for portfolios with stochastic pension goals and delegation. One could speculate that the reason for the wrong use of MPT to manage portfolios with stochastic goals and delegation to agents, and the potential looming crises, is that most papers that highlighted these nuances of GBI and agency were also piecemeal (e.g., Sharpe and Tint (1990) or Merton (2007) only focused on asset allocation for pension funds; Brennan (1993) only focused on asset pricing with agency). Practitioners, not well versed in the nuances of theory, did not adapt their practices despite theory having moved on from MPT to account for stochastic goals and agency. And theory continued to address these issues in a piecemeal fashion as this survey demonstrates. Hence, there is a pressing need to develop a new comprehensive theory that not only acknowledges the fact that investors attempt to achieve multiple stochastic goals (e.g., retirement, health savings, a child's education), and delegate to agents who they hope are skillful, but also provides the three facets (asset pricing, asset allocation, and riskadjusted performance) in a single, robust and consistent framework.

This paper will attempt to fill this void by offering another normative approach, the Goals and Risk-based Asset Pricing Model (GRAPM). It takes on this monumental challenge by focusing on a "pair-wise" equilibrium (i.e., using combinations of F, and two goal-replicating assets at a time) as opposed to a "one factor" equilibrium in MPT. Typically, academic theories have influenced investment practice, but this paper attempts to do the reverse and use investment practice to derive an academic theory of asset pricing, asset allocation, and risk-adjusted performance. In this multiple goal and agency model, 2FS is replaced by allocation to three assets, the asset pricing model depends on "relative" and not absolute betas, and the model appears to explain phenomena experienced in real life (e.g., why different investors can have different expected returns for the same asset; why seemingly similar investors-defined benefit pension funds-can have markedly different asset allocations). Unlike MPT, GRAPM is not dependent on (a) an unobserved risk aversion parameter, but rather on specific observed variables specified by institutional investors to both achieve their goal (e.g., absolute volatility targets) and the limits they place on agents (i.e., target tracking error); and (b) an unobserved market portfolio, but rather on goal-replicating assets that are potentially easily observed or created. Furthermore, the return of the absolute risk-free asset (which is risky for all goals) can be established endogenously in such a model. This approach provides an alternative heterogeneous investor model by assuming that each investor has a unique goal, and thereby has a unique asset that serves as the relative safe asset.

The paper is structured as follows: Section 1 provides a brief summary survey of the vast literature on agency and GBI and highlights which of the three facets of investing these key papers address.<sup>2</sup> It also briefly reviews the *M*-cube risk-adjusted performance measure of Muralidhar (2001) because risk-adjusted performance has had less prominence in the theoretical finance literature than asset pricing and asset allocation. Unlike the Sharpe, GH1, GH2 or *M*-square, *M*-cube—which is an extension of *M*-square—allows for both a stochastic goal and delegation. This is the measure of risk-adjusted performance

that will be used to derive the other two facets of investing and hence is critical to the paper. However, two key challenges stymie the search for and derivation of a new practical theory: the absence of "safe" assets for GBI, and the challenges of utility theory to address reality. Section 2 addresses the first issue and examines the unique nature of GBI and reviews recommendations to create goal-replicating assets, as these goal-replicating assets serve as the "safe" asset and benchmark for every goal. Section 3 addresses the second issue, noting that investors do not specify or maximize expected utility as assumed by theory, but rather maximize risk-adjusted returns.<sup>3</sup> Since utility theory has yet to incorporate multiple goals and agency into a single, simple function, this paper will use the Perold's (2004) approach of attempting to derive this model and the three facets of investing by maximizing relative riskadjusted performance using M-cube. Section 4 lays out GRAPM, which assumes that investors maximize goal relative risk-adjusted returns, subject to clearly articulated absolute and relative risk budgets and the desire to hire skillful agents. It explores the implications for pricing, managing and evaluating assets. GRAPM exploits the fact that a relative safe asset for one goal is risky for another (and vice versa), but every asset can have just one price/return. The equilibrium articulated in this section is a "pair-wise" equilibrium and thereby a departure from previous models. Adding more goals just requires more pair-wise equilibria (creating a "lattice" of equations). Section 5 highlights shortcomings and extensions and Section 6 summarizes.

#### 2 Summary of Literature on GBI and Agency and the Three Facets

When one includes GBI and agency, the paradigm changes from a world focused on absolute wealth, to one focused on relative wealth (Muralidhar, 2019b), and all investment aspects are influenced by the goal and/or the agent's benchmark.<sup>4</sup> Table 1 lists<sup>5</sup> major papers in finance that touch on GBI and agency, noting key contributions to theory (column 2), and categorizes them by two key criteria: first, whether the model is focused on absolute (column 3) or relative wealth, further broken down by type of stochastic goal, where applicable (columns 4-6); and second, which of the three facets of investing they provide insights on (columns 7-9). Column 10 provides additional comments, especially whether the papers are based on equilibrium or optimization models. For example, Markowitz (1952) is clearly focused on absolute wealth (with an "X" in column 3) and optimal asset allocation (column 8); examples of models on relative wealth (with an "X" in columns 4-6) include those focused on agency, background risk, peer comparison, liabilities, etc. All "relative models" implicitly or explicitly draw inspiration from Merton (1973), row 6, which while focused ostensibly on absolute wealth, is an equilibrium relative model. The stochastic opportunity set in Merton (1973) requires investors to implement Three Fund Separation (3FS); namely, that the portfolio should be split among the absolute risk-free asset, F, the risky portfolio, Mor M' depending on the circumstance, and the "hedging portfolio", L.<sup>6</sup> This crucial insight is the basis of all three facets of investing in the relative world.

The key asset pricing, asset allocation and riskadjusted performance conclusions of this extensive body of research can be summarized as follows:

*Asset Allocation*: All relative papers (e.g., rows 7, 8. 9. 13, 14, 15, 16, 21, 22 and 23) follow some form of 3FS as noted earlier. The only difference in the various papers is that the goal-hedging portfolio in goals-based papers (e.g., rows 7, 19, 21, 22, 23) changes to the benchmark replicating portfolio in agency-based papers (e.g., row 9). In other

Pa No. 1 Markowitz 2 Tobin (195: 3 Sharpe (19 4 Treynor (19 5 Black (197 6 Merton (19 6 Merton (19 7 Sharpe and 8 Abel (1990 8 Abel (1990 9 Brennan (1 (2005); 10 Sharpe (19 11 Graham an 12 Modigliani	per/author (Year) (1) (1952) 8) 64), Lintner (1965), (1966) 966) 973) 17int (1990) 0) 17int (1990) 0) 17int (1990) 17int (1990) 17int (1990) 17int (1990) 10	Key contribution (2) Mean-variance optimization Two Fund Separation Asset Pricing Model Treynor ratio Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	Absolute One deterministic si		Relative					
Pa No. 1 Markowitz 2 Tobin (195: 3 Sharpe (197 5 Black (197 6 Merton (19 6 Merton (19 7 Sharpe and 8 Abel (1990 8 Abel (1990 9 Brennan (1 (2005); 10 Sharpe (19 11 Graham an 12 Modigliani	per/author (Year) (1) (1952) 8) 64), Lintner (1965), (1966) 966) 973) 773) 17int (1990) 0) 993); Cornell and Roll Pressnan and Li (2008)	Key contribution (2) Mean-variance optimization Two Fund Separation Asset Pricing Model Treynor ratio Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	One deterministic si goal	,						
1         Markowitz           2         Tobin (195is)           3         Sharpe (19)           4         Treynor (15)           5         Black (197)           6         Merton (19)           7         Sharpe and           8         Abel (1990)           9         Brennan (1           (2005);         10           11         Graham an           12         Modigliani	(1952) 8) 64), Lintner (1965), (1966) 966) 973) 773) 1Tint (1990) 1) 993); Cornell and Roll 993); Cornell and Roll	Mean-variance optimization Two Fund Separation Asset Pricing Model Treynor ratio Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	(3)	One N tochastic st goal (4)	Multiple tochastic goals (5)	Other stochastic benchmark (6)	Asset pricing a (7)	Asset ullocation (8)	Risk- adjusted performance (9)	Comments (Equilibrium or Optimal Model) (10)
<ul> <li>2 Sharpe (190</li> <li>3 Sharpe (190</li> <li>4 Treynor (1)</li> <li>5 Black (197</li> <li>6 Merton (19</li> <li>7 Sharpe and</li> <li>8 Abel (1990</li> <li>8 Abel (1990</li> <li>9 Brennan (1</li></ul>	00) (1966) (1966) (1966) (1966) (17) (17) (17) (1990) (17) (1990) (1993); Cornell and Roll (1993); Cornell and Roll (1993); Cornell and Roll	Asset Pricing Model Asset Pricing Model Treynor ratio Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	×					x ×		Risky market portfolio—Optimal
4         Treynosun           5         Black (1971           6         Merton (19           7         Sharpe and           8         Abel (1990           9         Brennan (1           (2005);         10           10         Sharpe (19           11         Graham an           12         Modigliani	(1200) 366) 773) 1Tint (1990) 993); Cornell and Roll Pressnan and Li (2008)	Treynor ratio Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	x				x	<		CAPM—Equilibrium
<ul> <li>5 Black (197, 5)</li> <li>6 Merton (19</li> <li>7 Sharpe and</li> <li>8 Abel (1990)</li> <li>8 Abel (1990)</li> <li>9 Brennan (1 (2005);</li> <li>10 Sharpe (19</li> <li>11 Graham an</li> <li>12 Modigliani</li> </ul>	2) 7(3) [Tint (1990) 0) 993); Cornell and Roll Bessman and I i (2008)	Zero beta CAPM Intertemporal CAPM Liability focused asset allocation	Х	;			X		Х	Rating mutual funds—Equilibrium
<ol> <li>7 Sharpe and</li> <li>8 Abel (1990</li> <li>9 Brennan (1 (2005);</li> <li>10 Sharpe (19</li> <li>11 Graham an</li> <li>12 Modigliani</li> </ol>	(Tint (1990) )) 993); Cornell and Roll Pressnan and I i (2008)	Liability focused asset allocation	Х	X	0	Apportunity set	××	Х		Non-zero volatility of r(F); Equilibrium Continuous Time Model; 3FS;
<ul> <li>8 Abel (1990</li> <li>9 Brennan (1 (2005);</li> <li>10 Sharpe (19</li> <li>11 Graham an</li> <li>12 Modigliani</li> </ul>	) 993); Cornell and Roll Premium and Li (2008)			X				X		Equilibrium Allocation for nensions—Ontimal
<ol> <li>Brennan (1) (2005);</li> <li>(2005);</li> <li>10 Sharpe (19</li> <li>11 Graham an</li> <li>11 Graham an</li> <li>12 Modigliani</li> </ol>	993); Cornell and Roll Brannan and I i (2008)	reeping up with jones unity		1		Habit	x			Relative utility to single
(2005); 10 Sharpe (19) 11 Graham an 12 Modigliani	Pronnan and Li (2008)	Agency	Х			Agency	Х			Agency relative asset
<ol> <li>Graham an</li> <li>Modigliani</li> </ol>	<u>שוווומוו מווע בו (בעעער)</u> 94)	Sharpe and differential Sharpe ratio	Х			Agency			Х	pricing—Equilibrium Differential Sharpe =
<ol> <li>Graham and</li> <li>Modigliani</li> </ol>						0		;		Agency—Equilibrium
	d Harvey (1994, 1997) and Modioliani (1997)	GHI and GH2 <i>M</i> -square risk-adiusted	××			Agency Agency		××	××	Related to Black (1972)—Optimal Performance of mutual
		performance	1			fama ar		1	1	funds—Optimal
13 Heaton and	1 Lucas (2000)	Background risk			В	ackground risk	X			Alternative stochastic
14 Muralidhar	.(2001)	<i>M</i> -cube risk-adjusted performance		×		Agency		X	Х	benchmark—Equilibrium Ranks agents based on skill—Optimal
15 Lauterbach	and Riesman (2002,	Relative preferences—KUJ		1		Peers	×	1	1	Relative (to peers) asset
2004)										pricing—Equilibrium
16 Gómez and	I Zapatero (2003)	Two-factor CAPM	>		-	Benchmarking	×	>	>	KUJ Utility usage—Equilibrium
1/ Ferulu (20) 18 Merton (20)	(10)	Focus on funded status	<	X			<	<	<	Safe MPT asset is risky for GBI
19 Waring and	ł Whitney (2009)	Asset liability asset pricing		х			x			Black (1972) with stochastic
										liability—Equilibrium
20 Das et al. (	2010)	Mental accounts			X			x		Optimal allocation with multiple
10 Muralidhar	and Shin (2013)	Heina M-cuba with a acal and		X		Agency		*	Λ	goals—Optimal Belativa A seat Driving Model v
		osing int-cube with a goal and agency		<		nguiry		<	<	0.1—Optimal
22 Muralidhar	: et al. (2014a, 2014b)	Relative asset pricing model		Х			x			Single stochastic goal-Equilibrium
23 Deguest et	al. (2015)	Goals-based investing		Х			х			Asset allocation for GBI—Optimal
24 Das et al. (	2018)	Goals-based investing			Х			х		Joint optimization of goals-Optimal
25 This paper	(2019)	Goals and Risk-Based Asset Pricing Model			Х	Agency	X	x	x	Following Perold (2004)-Equilibrium

SECOND QUARTER 2021

JOURNAL OF INVESTMENT MANAGEMENT

#### ASSET PRICING, ASSET ALLOCATION AND RISK-ADJUSTED PERFORMANCE WITH MULTIPLE GOALS AND AGENCY

93

Not for Distribution

words, for a specific goal like retirement, Sharpe and Tint (1990), row 7, note that the hedging portfolio would be the portfolio that replicates the cash flows of the pension fund; for agency on the other hand, Brennan (1993), row 9, notes that it would be the benchmark (without further description of what the optimal benchmark should be). When one combines GBI and agency, Muralidhar and Shin (2013), row 21, argue that in this situation common among institutional pension funds, the optimal benchmark asset for agents should be the same as the goal-replicating asset. While this survey uses a broad "relative" umbrella, there are some interesting differences between asset allocation recommendations for GBI and agency.

Merton (2007), row 18, argues that retirement or GBI investors should focus on maximizing (expected utility of) funded status and not wealth. As a result, if the investor is fully funded or overfunded (i.e., has sufficient assets to cover their goal), then the goal is ideally completely hedged by allocating to the goal-hedging portfolio. Interestingly, in the late 1990s, many US public pension funds were overfunded (Aubry et al., 2018), but continued to use MPT to set allocations. Investing in assets deemed very risky relative to the goal, these pension funds face major challenges after two major equity market corrections. Merton (2007) characterized this investment behavior as "mispriced risk". If investors are underfunded as assumed in Sharpe and Tint (1990) and Muralidhar et al. (2014a), and is widely the case globally in 2020, then the allocation to the three assets, F, L and the risky portfolio will depend on the degree of underfunding and risk aversion. Typically, the greater the underfunding, the greater the allocation to risky assets.<sup>7</sup> This pattern of behavior has been documented in IMF (2013) and Aubry and Crawford (2019)—post-2008, even though equity markets have rallied, US public pensions have increased their allocation to risky assets as funded status declined.

In the case of pure agency, the allocation to the risky asset is determined by the relative risk permitted by the principal. In agency papers, because agents want to maximize relative return to get compensated, they seek as much relative risk as possible. The principal fears they are not skillful (and would gamble the money) and hence seeks to limit relative risk. However, in both GBI and agency, the relative risk taken/permitted reflects the risk tolerance of the investors and is easily observed, either in an investment policy statement or the actual portfolio positions (discussed in Section 3).

Asset Pricing: In all the relative models noted in Table 1 (e.g., rows 7, 8. 9. 13, 14, 15, 16, 21, 22 and 23), F exists with zero net supply, and has an exogenous return. The "hedging portfolio", L, because of its hedging property, is the "relative" risk-free asset and will typically earn a return below that of the risky (relative) market portfolio. Muralidhar et al. (2014a), row 22, make an interesting distinction regarding the "market" portfolio. They argue that in a single stochastic goal model, F is risky relative to L. As a result, the third portfolio, M' (and not the traditional Min CAPM or even Merton, 1973), is a unique, risky relative market portfolio. M' includes Fand excludes L and, like CAPM, also includes all risky assets being priced. This approach creates complications in a multiple goal model as each goal has its own relative risky market portfolio, making an overall equilibrium extremely hard to solve.

Consider a Relative Asset Pricing Model (RAPM), which can be seen as a proxy for all "relative" models. Equation (1) highlights the basic asset pricing model in this relative paradigm. The intuition and the basic setup of the RAPM is provided in Appendix 1. Since it follows many of the steps used to derive CAPM, its form is similar to the CAPM model with a few variations.

$$E[r(i) - r(F)] = \frac{\operatorname{cov}[r(M') - r(L), r(i)]}{\operatorname{var}[r(M') - r(L)]} \times E[r(M') - r(L)] = \Gamma_{M'-L,i} \times E[r(M') - r(L)]$$
(1)

where  $\Gamma_{M'-L,i}$  = is a "relative beta", which is a relative covariance term divided by a relative variance term. [r(M') - r(L)] is the relative (to goal) market risk premium. Gómez and Zapatero (2003), row 16, term this relative asset pricing model (in their specific example) a "Two-Factor CAPM," because the pricing model depends on M' and L. For simplicity, we use RAPM as a generic reference for all relative models in Table 1 as they all largely follow Equation (1). "L" could easily be the peer benchmark (row 8), the benchmark to which an agent is measured by the principal (row 9), a background risk process (row 13), the home bias benchmark (row 15), or any other relative variant listed above (or even Campbell and Cochrane (1999) focused on habit). In this model, much like CAPM, asset I is a constituent of M'.

MPT can be seen as a very specific case of RAPM—with an implicit assumption that the goal is deterministic. If liabilities are deterministic (i.e., r(L) = r(F)), CAPM is retrieved. Under this assumption, maximizing absolute and relative wealth are identical objectives.<sup>8</sup> RAPM, similar to CAPM, makes a strong assumption and assumes that all investors have the same identical stochastic goal or benchmark. The challenge with RAPM is that "L" is not clearly specified in practice, as each investor has a unique L. Hence, it is not easy (and probably impossible) to test RAPM empirically, though the 3FS

recommendations for asset allocation are optimal for individual investors and often observed in innovative/advanced pension, insurance, and endowment portfolios.

F and L are critical elements of asset pricing, asset allocation, and risk-adjusted performance in a relative world because: (i) F has zero volatility and zero correlation to all other assets including L and M'; and (ii) L has zero relative risk relative to itself (or perfectly correlated to the goal), but is otherwise like any other asset. This unique assumption of zero absolute volatility, zero correlation, and zero relative volatility embedded in Fand L is critical to establishing risk-adjusted performance measures in a goals- and agency-based world.

*Risk-Adjusted Performance*: The simplest (and zero effort or zero skill) way for an agent to outperform a benchmark is to borrow through F and lever the benchmark to which they are measured. This delegated portfolio will outperform the benchmark on average, and be rewarded unless the principal appropriately risk-adjusts performance for the higher volatility of the levered portfolio. Modigliani and Modigliani (1997) normalize the performance and volatility of an asset (or agent) by leveraging/deleveraging this risky asset, using F, to achieve a target volatility (typically of the benchmark). The M-square risk-adjusted performance clearly nullifies naïve leverage used by an agent to beat a benchmark.

Assume that a principal has a goal that is replicated by L and hires an agent, who creates an active portfolio P. The principal cannot observe the true effort of the agent and can only monitor P's returns. The principal wants to maximize the relative risk-adjusted return of the portfolio (through actions of their own and the agent's portfolio), but does not want to pay the agent for zero "intelligence" activities like leverage or "beta". The M-square performance of the agent can be established by the principal using two conditions shown in Equations (2) and (3). r(\*) measures the return of an asset or portfolio.

$$r(A) = d \times r(P) + (1 - d) \times r(F) \qquad (2)$$

subject to  $\sigma(A) = \sigma(L)$  (3)

The principal first calculates the "d" for each agent, where  $d = \sigma(L)/\sigma(P)$ ,  $\sigma$  is the volatility, and d measures the implicit or explicit leverage in portfolio P. d greater than 1 indicates the agent levered the portfolio; d less than 1 indicates deleverage.<sup>9</sup> The principal then calculates risk-adjusted return, r(A), for each agent to rank them after levering or delevering portfolio *P* to have the same volatility as the benchmark, L. Since all agents have the same absolute risk, their *M*-square returns (or r(A)s) of all agents are now comparable. M-square ranks agents identical to the Sharpe ratio, but is expressed in terms of returns (as opposed to a ratio) and provides insight into how much of the return is derived from naively levering/delevering the benchmark. One unique *M*-square insight is than an agent with a portfolio that underperforms L (i.e., r(P) <r(L), might actually be quite talented and should be hired if their r(A) exceeds r(L). In the traditional world of investing, this agent might be seen as an underperformer and not selected. But Msquare includes allocation advice; namely, lever this manager's returns to have the same volatility of the benchmark as that combination outperforms the benchmark. This is a non-equilibrium model so it is possible for r(A) s generated by portfolios of different agents to have different returns, even though they all have the same volatility (because correlation to the benchmark has not been normalized).

Muralidhar (2001) extends this approach by arguing that in a "relative" paradigm, especially agency, there is a dual measure of risk: both absolute and relative risk (or tracking error—TE hereafter), and not just absolute risk as in *M*-square or MPT. Therefore, an agent's performance P must be risk-adjusted, by levering/delevering using both F and the benchmark hedging asset, L, to create a new portfolio with both a target absolute volatility ( $\sigma(L)$ ), as in *M*-square, and a target correlation to the benchmark (or de facto, target relative risk). The agency literature has ignored this second constraint of normalizing correlations as well and hence typical measures cannot rank agents identical to ranking based on measures of confidence in skill (Ambarish and Seigel, 1996; Muralidhar, 2001). This is a very important point as principals would ideally want only skillful agents. M-cube ensures that all agents' risk-adjusted performance (r(A)s) have the same relative risk (TE), but also normalized to have the same volatility ( $\sigma(L)$ ) and target correlation. This is also a non-equilibrium model like *M*-square.

Much like *M*-square, *M*-cube is estimated by assuming that principals maximize the goal relative risk-adjusted return, r(A) - r(L), subject to three constraints highlighted in Equations (4)–(6).

$$r(A) = a \times r(P) + l \times r(L)$$
  
+  $(1 - a - l) \times r(F)$  (4)

$$\sigma(A) = \sigma(L) \tag{5}$$

and 
$$TE(A) = TE(Target)$$
 (6)

where *a* is the allocation to the risky (agent) portfolio, *P*, and *l* is the allocation to the goalhedging portfolio, *L* (and measures what is *de facto* invested in the low-cost passive benchmark or "beta" in industry parlance). The balance of the assets is invested in the traditional risk-free asset, *F* (and measure leverage). The term "1 - a - l" is the corollary to 1 - d in *M*-square.

This approach is described in more detail in Appendix II as the derivation involves a few more steps than *M*-square. To summarize, if  $\rho(P,L)$  is the correlation of returns of *L* and *P* of each agent (which is easily calculated), and  $\rho(T, L)$  as

the target correlation,<sup>10</sup> which is specified by the principal, we can solve for "*a*" as in Equation (7), and "*l*" in Equation (8). The super-script denotes the asset being allocated to; the subscript indicates the target relative risk (T) and the goal-replicating asset (L).

$$a_{T/L}^{P} = \frac{\sigma(L)}{\sigma(P)} \left[ \sqrt{\frac{[1 - \rho(T, L)^{2}]}{[1 - \rho(P, L)^{2}]}} \right]$$
$$= \frac{\sigma(L)}{\sigma(P)} \left[ \frac{\varphi(T, L)}{\varphi(P, L)} \right], \quad \text{where } \varphi(I, J)$$
$$= \sqrt{[1 - \rho(I, J)^{2}]} \tag{7}$$

$$l_{T/L}^{L} = \rho(T, L) - \rho(P, L)$$

$$\times \left[ \sqrt{\frac{[1 - \rho(T, L)^{2}]}{[1 - \rho(P, L)^{2}]}} \right]$$

$$= \rho(T, L) - \rho(P, L)$$

$$\times \left[ \frac{\varphi(T, L)}{\varphi(P, L)} \right]$$
(8)

"a" and "l" provide the allocation to risky and the relative risk-free assets in a non-equilibrium but optimal model, respectively, for a given T, L and P. If the investor sets the target correlation  $(\rho(T, L))$  to the correlation of the agent's portfolio to the goal ( $\rho(P, L)$ ), then the solution to "a" is "d" in M-square and "l" = 0. In other words, because MPT assumes that the investor is principal (and ignores goals), it tolerates whatever relative risk is actually taken in the portfolio and is insensitive to relative risk. Equations (7) and (8) also highlight the factors that will drive the demand for/allocation to these assets (discussed in Section 3). Very simply, the allocation to risky assets is positively impacted by the volatility of the goal and (non-linearly) by the target TE. The allocation to the goal-hedging asset is invariant to volatilities, but is inversely impacted

by the target TE—lower the target TE, the more likely the investor is to increase the hedge (see also Appendix III). By contrast, in traditional MPT, the allocation to an asset is determined by either its market capitalization and linearly to the allocation to risky assets because of risk aversion (in the equilibrium CAPM model because of 2FS) or by an opaque calculation based on whether it is included in an efficient portfolio or not (Markowitz, 1952). It is unaffected by potential substitutes and complementary assets. Section 3 will demonstrate how Equations (7) and (8) appear to provide allocations not very different from actual allocations of a large institutional investor, with similarly articulated risk targets.

Muralidhar (2001) shows how M-cube ranks agents consistent with measures based on confidence in skill (Ambarish and Seigel, 1996) and hence is preferable to Sharpe, GH1/GH2 and Msquare in agency arrangements. This is the case because Sharpe, GH1/GH2, and M-square only focus on normalizing volatility (i.e., absolute risk) and ignore correlation (which impacts relative risk).<sup>11</sup> In short, agents with portfolios naively levered or highly correlated to the benchmark are shown to have lower risk-adjusted performance (and lower confidence in skill) than agents with truly unique portfolios. After all, generating excess performance (over the benchmark) by naïve leverage using F or allocating to L does not require skill and should not be rewarded.

Table 2 encapsulates the research reviewed and complements it with examples from practice to summarize the implications of these findings for a new theory and practice. It highlights the differences in behaviors when one compares Traditional MPT (absolute wealth and principals) with Investment Reality (multiple goals and agency). MPT assumes certain behaviors; behaviors in reality are easily observed and hence this

	Traditional MPT	Investment reality
Assumed vs observed behavior	Assumed	Observed
Objective	<ol> <li>Maximize expected utility of wealth by generating highest return per unit of risk</li> </ol>	<ol> <li>Invest to service many goals (retirement; college expenditures; health).</li> <li>Investors (should) care about funded status or relative wealth</li> </ol>
Decision maker	1. Individual (principal)	<ol> <li>Institutional boards (principals) delegate to investment staff (agents), who then delegate to external managers (agents). Includes pension funds, endowments, insurance companies, sovereign wealth funds with different goals and governance structures.</li> </ol>
	2. Single investor type	<ul><li>2. Retail investors (principals) delegate to advisors (agents), who then delegate to mutual fund or other managers (agents)</li></ul>
Risk parameter	1. Volatility of portfolio	<ol> <li>Absolute risk of the goal often stated in Investment Policy Statements (IPS)</li> <li>Relative risk of portfolio delegated to agents; often stated clearly by institutional investors in IPS</li> <li>Drawdown should matter more than volatility because it impacts compounding, but few investors use it explicitly in IPS statements</li> </ol>
Risk aversion parameter	1. Critical to the model	<ol> <li>Correlation of the active portfolio to the benchmark;</li> </ol>
	2. Exogenous variable never observed in practice	2. Easily observed and measured;
Skill	Not a concern because principal does not delegate or pay fees	A huge concern because principals do not want to pay fees for luck

 Table 2 Traditional MPT vs Investment Reality—Implications for theory and practice.

contrast is useful. Given the substantial differences between Traditional MPT and Investment Reality, it is critical to establish an effective theory based on reality.

#### **3** Demand and Supply of Absolute and Relative Risk-Free Assets

While the thorny CAPM question has been the identification of M, the key question for GRAPM that must be addressed before deriving or even suggesting a new model is whether assets F and L even exist? After all, L in RAPM (Equation (1)) was not clearly identified and articulated, and hence while potentially theoretically interesting, it is an impractical model to estimate asset prices. The principal challenge to adopting GBI (and as a result 3FS) in reality or in articulating a new theory is that there are many goals-each goal is unique, and often the goal-replicating asset does not exist. Sharpe and Tint (1990) demonstrate how one might proxy the cash flows of a defined benefit pension fund (which aggregates the liabilities of many individuals) or an insurance company. Das et al. (2010), among the first to examine a multiple goals approach, develop a (mental accounts) asset allocation recommendation by assuming that every goal has a target expected return. However, this assumption also potentially misses the complexity of current GBI. For example, a 25-year old wishing to save for retirement or a new mother wishing to save for their newly born daughter's college fund cannot buy a liquid, traded instrument today to hedge the cash flows of these very realistic goals. Clearly, one should not derive asset allocation recommendations or asset pricing models for a host of goals based on assets that do not exist. Fortunately, recent work has sought to address this challenge and potentially clears this obstacle from the path to a new theory.

Many investment goals, including retirement, college savings, or buying a house typically, have one feature in common-namely, they can be specified as a set of forward-starting real cash flows (certain or uncertain), indexed to some appropriate measure of inflation, which then last for a (certain or uncertain) period of time once they start. They are clearly not deterministic. Muralidhar et al. (2016), and Merton and Muralidhar (2017a), suggest that the goal of retirement is to receive a target, guaranteed, real income, from retirement, say age 65, to death at say 85. In short, individuals save for approximately 40 years, and invest these savings in order to receive these specific cash flows to allow them to have an adequate retirement lifestyle.<sup>12</sup> Similarly, we make the assumption that the goal of saving for a child's college education is to be able to receive a targeted, guaranteed, real income (to pay for tuition and other costs), starting when the child is 18 years old, for a minimum period of 4 years. Here the link would be to tuition inflation.

The key contribution of Muralidhar *et al.* (2016) and Merton and Muralidhar (2017a) is to argue for the creation of a new bond by governments, or even private institutions like insurance companies, that mimics the desired retirement cash flows-Muralidhar et al. (2016) call these Bonds for Financial Security (BFFS) and Merton and Muralidhar (2017a) call them Standard-of-Living Forward-starting Income-only Securities (SeLF-IES) and they differ only in the inflation index to which they are linked.<sup>13</sup> Similarly, Muralidhar (2016a, 2016b) argue for the creation of Bonds for Education and Student Tuition (BEST) that would mimic the college cash flows, and would be issued by colleges and universities. These goalappropriate bonds would serve as the relative risk-free asset for these specific goals, greatly simplifying GBI (Muralidhar, 2016b, 2019a, 2019b), and there is an easily identified issuer who

would benefit from issuing these bonds (governments for SeLFIES; colleges for BEST), thereby ensuring supply.

Proceeding in a similar fashion, we can make the assumption that every stochastic asset (or combination of assets) that exists potentially meets a unique goal, and obviously there has been a natural issuer who has sought to provide the instrument. Where instruments do not exist to hedge current stochastic goals for say health savings accounts (i.e., where there is demand), the hope is that finance science could potentially lead to the issuance of the new instrument by a supplier as noted in the case of two major goalsretirement and saving for college.<sup>14</sup> In summary, we assume that for each goal (or L) today, either the instrument exists and is currently traded, or finance science needs to be used to create it as there clearly is demand.

But what about F? In MPT/CAPM, two interesting assumptions are made about F, the "safe" or (absolute) risk-free asset to make the model work: (a) that F exists with net zero supply; and (b) that the rate of return is exogenously determined. These are clearly strong assumptions. In practice, investors use the Treasury-bill as a proxy for F, but clearly, the rate of return of T-bills is determined by market forces and it has positive supply. But in a multiple goals-based world, it may be reasonable to assume that there is a unique goal, for some individual, with the cash flow profile of the absolute risk-free asset, F (i.e., there is a traditional MPT investor in this world). The goal replicated by F would be deterministic and would be for an investor that is absolute wealth focused. And similarly, there exists a natural supplier of such instruments (e.g., governments) who benefit from issuing such bonds. In short, this (absolute) risk-free asset, along with all other assets, could exist with positive net supply and an effective model could establish its return endogenously based on market-clearing conditions. Interestingly, this approach of assuming that: (a) every existing asset could possibly satisfy a unique goal of some individual in the economy, or (b) assets could be created for goals without current replicating assets by potentially a natural issuer, is another approach to developing a "heterogeneous investor" model for asset pricing.<sup>15</sup> Section 4 will show how the multiple goal setting could potentially dispense with the need to identify and establish *M* as long as *F* and at least two goal-replicating assets exist.

#### 4 Maximize Expected Utility or Risk-Adjusted Performance?

The second obstacle to using the traditional approach to derive a new theory that captures multiple goals and agency is that finance theory is largely based on the assumption that investors maximize expected utility (of wealth). While relative utility functions have been used to address a single goal or agency as shown in Table 1, there is no easily tractable (or currently specified) utility function that could handle (multiple) goals and agency within a single construct. Moreover, Equation (1) for RAPM suggests that each goal has its own unique relative market portfolio; hence an equilibrium with many goals would make for a very complex set of allocation formulas across these many relative market portfolios. More critically, as Table 2 has noted, investors do not maximize expected utility—instead they maximize relative to goal, risk-adjusted, expected returns.

For example, two large and innovative pension funds (the Los Angeles County Employees' Retirement Association—LACERA—and the New Mexico Public Employees' Retirement Association—NMPERA) have very explicit statements of objectives that are worth reviewing. The LACERA Investment Policy (IPS) states: "The Fund's long-term performance objective is to generate risk-adjusted returns that meet or exceed its defined actuarial target as well as its policy benchmark, net of fees, over the Fund's designated investment time horizon." NMPERA's IPS explicitly states that the Board established a 10.5% annualized target volatility for the strategic asset allocation (or L) and a 1.5%annualized tracking error (or TE(Target)) for all delegated decisions.<sup>16</sup> LACERA, like NMPERA, also articulates a relative risk budget. These are not random examples, but reflective of the general practice among many sophisticated investors. More importantly, when one applies Equations (7) and (8) for these given absolute (10.5%) and relative risk parameters (1.5%), M-cube suggests a 63% allocation to risky assets and a 37% allocation to the relative risk-free asset-which is not very different from the actual NMPERA allocations.<sup>17</sup>

Muralidhar (2019b) suggests that finance theory should restate the original six-word phrase, "individuals maximize the expected utility of wealth" as "individuals/institutions delegate to maximize risk-adjusted relative-returns". The latter approach is much more reflective of how investors actually behave (Table 2); namely, they maximize wealth relative to their goals (i.e., relative wealth matters and these goals are very different from one another), and they delegate to agents (i.e., relative risk matters). More importantly, investors do not specify utility functions (Markowitz, 1990), but rather seek to maximize risk-adjusted relative returns as clearly shown in the LACERA IPS. As a result, the Perold's (2004) approach is adopted hereafter, to derive the three facets in a multiple goals and agency world. The *M*-cube measure lends itself to this approach because it can be used for both goals (i.e., L) and agency (i.e., TE(Target)) in one formula.

#### 5 The Simple Goals- and Risk-Based Asset Pricing Model

With two goal-specific relative risk-free assets and the traditional (absolute) risk-free asset, and investors attempting to maximize risk-adjusted returns based on *M*-cube, it appears that all assets can be priced. This model does not require a "market portfolio" because the approach uses a "pairwise equilibrium," and a generic risky asset can be priced with any goal-replicating asset (which is risky for another goal). This section is in the spirit of Debreu (1959), Arrow (1964), Breeden and Litzenberger (1978), and Cochrane (2005), but based on goals, risk budgets, and goal-replicating assets as opposed to Arrow-Debreu securities or options prices on aggregate consumption or stochastic discount factors.<sup>18</sup> The intuition behind this model is very simple: a relatively risk-free asset for one goal (e.g., SeLFIES for retirement) is a risky asset for another goal (college savings) and vice versa. Equation (7) shows that the goal-replicating asset, F and any risky asset can be used to create risk-adjusted portfolios with specific target volatility and correlation characteristics to the goals. If such portfolios can be created for the first goal (first using a generic risky asset and then the second goal-replicating asset), the equilibrium condition for two such portfolios with identical risk characteristics for that goal gives one key equation to price a generic risky asset. But the same is true for the second goal giving a second pricing equation for the generic risky asset. Free trading across investors with different goals forces a "pair-wise" equilibrium as an asset can have only one price/return despite having different attributes (e.g., substitute or complement) for each goal. Adding more goals and goal-replicating assets just adds more "pair-wise" equilibria.

Assumptions: The following assumptions are required:

(i) Assume a one-period world

- (ii) Investors/principals have a goal and delegate to a single agent to achieve that goal
- (iii) Principals seek the highest goal relative expected risk-adjusted return for their goal
- (iv) Agents take risk relative to the relative riskfree asset (either because the principal is underfunded,<sup>19</sup> is incapable of managing assets, or claim to have skill in outperforming the benchmark). Principals have limited relative risk budgets as they are uncertain about the skill of agents and can only observe past returns, not effort or skill. They extrapolate from past returns to forecast expected returns.
- (v) Principals specify risk targets as shown in Section 3—with a specific absolute risk level<sup>20</sup> equal to the risk (or volatility) of the goal, and a relative risk budget for their agents (or TE(Target))
- (vi) There are two classes of investors pursuing two different goals, G1 and G2. We assume a single representative investor for each goal. Initially, we assume that they are isolated and cannot trade, but then relax that assumption to ensure an equilibrium.
- (vii) Both these investors have unique goalreplicating assets, G1 and G2, respectively
- (viii) Assume F exists. Also, assume a generic risky asset I, that we are seeking to price and different from G1, G2, and F, exists.
- (ix) In other words, assume that there is sufficient supply of all assets (and ignore supply-side issues)
- (x) Assume agents can invest in just a single risky asset to create the delegated portfolio relative to the goal– either *I* or the second goal-replicating asset
- (xi) Principals can combine risky assets, with *F* and a goal-replicating portfolio to create risk-adjusted portfolios, *A*
- (xii) For both investors, the target correlation or  $\rho(T,L)$  is identical (not the TE(Target) as that also depends on the volatility of the

goal). This just simplifies the formulas, not the end result

- (xiii) There are no transaction costs and unrestricted trading ensures equilibrium
- (xiv) The investors have equal assets under management so that the initial simple model is free of the fraction of investors for each goal

Solution. Assumptions (i)–(iii) allow us to use the M-cube measure to derive the model. Thereafter, eight major steps are needed to establish the "pairwise" asset pricing equations.<sup>21</sup>

- (1) Recall Equations (7) and (8) establish optimal allocations to risky, goal-replicating and risk-free (*F*) assets for investors maximizing goal-relative risk-adjusted returns, given target absolute and relative risk levels. At this stage, this is an optimal, but non-equilibrium allocation.
- (2) First consider the G1 world in isolation. Use Equation (4) to establish the risk-adjusted return of portfolio *A*,  $(E[r(A)_{I,G1}])$ , in the G1 world with I as risky asset, and goal-replicating asset G1.
- (3) Similarly, establish the risk-adjusted return of portfolio *A*,  $(E[r(A)_{G2,G1}])$ , in the G1 world with G2 as risky asset.
- (4) This is the most important step and claim in this paper. If portfolio A has the same volatility as the goal (G1) and the same target correlation ( $\rho(T,G1)$ ), then it cannot have two values based on which asset was used to create A. Therefore, set  $E[r(A)_{I,G1}] =$  $E[r(A)_{G2,G1}]$  to derive  $E[r(I)_{G1}]$  as function of G1, G2, and F.<sup>22</sup>
- (5) Similarly, in the G2 world, follow Steps 1– 3 to derive  $E[r(A)_{I,G2}]$  and  $E[r(A)_{G1,G2}]$ . Once again, as in Step 4, set  $E[r(A)_{I,G2}] = E[r(A)_{G1,G2}]$  to derive  $E[r(I)_{G2}]$  as function of G1, G2, and *F*.

(6) Now assume the two investors can trade freely. Asset *I* cannot have two different returns unless there is a market inefficiency. Therefore, set  $E[r(I)_{G1}] = E[r(I)_{G2}]$  to express E[r(I)] as function of E[r(G1)], r(F), volatilities and correlations (assuming G2 is the goal). This is shown in Equation (9).

$$E[r(I) - r(F)] = Z_{I,G1/G2} \times E[r(G1) - r(F)]$$
(9)

where  $Z_{I,G1/G2}$ 

$$= \frac{\sigma(I)}{\sigma(G1)} \left[ \left[ \frac{\varphi(I, G2)}{\varphi(G1, G2)} \right] + \frac{Y_{I,G1,G2}}{X_{I,G1,G2}} \rho(I, G2) - \frac{Y_{I,G1,G2}}{X_{I,G1,G2}} \rho(G1, G2) \times \left[ \frac{\varphi(I,G2)}{\varphi(G1,G2)} \right] \right]$$
(10)

and where  $X_{I,G1,G2}$ 

$$= \varphi(I, G1) - \rho(I, G2)$$

$$\times \varphi(I, G1) + \rho(G1, G2)\varphi(I, G2)$$
(11)

 $Y_{I,G1,G2}$ 

$$= \varphi(I, G2) - \rho(I, G1) \times \varphi(I, G2)$$
$$+ \rho(G1, G2) \times \varphi(I, G1)$$
(12)

Zeta can be restated in terms of "a" and "l" using Equations (7) and (8) as follows:

$$Z_{I,G1/G2} = \frac{a_{T/G2}^{G1}}{a_{T/G2}^{I}} + \frac{\sigma(I)}{\sigma(G1)} \times \left(\frac{Y_{I,G1,G2}}{X_{I,G1,G2}}\right) l_{I/G2}^{G2} \quad (13)$$

(7) Similarly, express E[r(I)] as function of E[r(G2)], r(F), volatilities, and correlations (assuming G1 is the goal) as in Equation (14).

$$E[r(I) - r(F)] = Z_{I,G2/G1} \times E[r(G2) - r(F)] \quad (14)$$

(8) *F* has an equilibrium value because of the unique role it plays in risk-adjusted portfolios and can be calculated as

$$r(F)_{G1,G2} = \frac{\begin{cases} E[r(G1) - \frac{\sigma(G1)}{\sigma(G2)} \\ \times (\frac{X_{I,G1,G2}}{Y_{I,G1,G2}}) \times E[r(G2)] \\ \end{cases}}{[1 - \frac{\sigma(G1)}{\sigma(G2)} \times (\frac{X_{I,G1,G2}}{Y_{I,G1,G2}})]}$$
(15)

(9) Additionally, there is a unique relationship between E[r(G1)] and E[r(G2)] as in Equation (16).

$$E[r(G1) - r(F)] = \frac{\sigma(G1)}{\sigma(G2)} \times \left(\frac{X_{I,G1,G2}}{Y_{I,G1,G2}}\right) \times E[r(G2) - r(F)]$$
(16)

*Implications of GRAPM*. There are many interesting implications from this approach and model.

- (i) GRAPM incorporates the three facets into one model (an asset pricing model, asset allocation and *M*-cube as the risk-adjusted performance measure), while incorporating the realities of investing: multiple goals, delegation and precise, observable risk specification.
- (ii) A generic asset I can be priced with just a goal-replicating asset G1 (Equation (9)) or G2 (Equation (14)). These assets, G1 and G2, can be observed as opposed to an unobserved absolute (M) or relative market portfolio (M').

- (iii) There are potentially multiple equations to price a particular asset depending on the number of goal-replicating assets. This happens because of the pair-wise equilibria. This is not a shortcoming of the model, but rather a feature and is visualized in Figure 1. While CAPM and RAPM provide a single equation linked to equilibrium market portfolios (which include the generic risky asset) and goal-replicating portfolios, GRAPM provides multiple pair-wise equilibrium formulas based on the number of goal-replicating assets. With two assets, G1 and G2, I can be priced with G1 as the goal and G2 as risky, or vice versa. With three assets, I can be priced using six combinations of G1, G2, and G3. The threegoal case yields some additional intriguing equilibrium conditions that will be addressed in future research (Muralidhar, 2020).
- (iv) Alternatively, if there are other conditions that prevent an equilibrium, one can expect that we can derive a range of expected returns for each asset—a more reasonable assumption than having a single global point estimate (assuming everyone has the

same expectation). Interestingly, when one compares the forecasts of say consultants who advise institutional or retail clients, there can be wide dispersion in their forecasts and one explanation is that they are only looking at the forecast from the bias of their goals and not from a global equilibrium that Equations (9) and (14) capture. Furthermore, forecasts of expected returns have been poor both in terms of level and direction (Housel, 2015; The Economist, 2017) and hence potentially having a range of forecasts may be more valuable for practitioners.<sup>23</sup>

- (v) F cannot take an arbitrary, unspecified value but rather as a very specific value as shown in Equation (15). The equilibria force this result.
- (vi)  $X_{I,G1,G2}$  (Equation (11)) and  $Y_{I,G1,G2}$  (Equation (12)) are only functions of the correlation parameters. This is also the equilibrium condition between r(G1) and r(G2); namely, that in equilibrium, given the unique relationship of these goal-replicating assets, their values will also be determined endogenously in the market and cannot take any exogenous value.



Figure 1 Asset pricing relationships for CAPM, RAPM, and GRAPM (with two and three goals).

- (vii) The Zeta term is in the spirit of the beta term of MPT, and relative beta of RAPM. It is a "covariance" term with many additional correlation terms to capture the equilibrium conditions described above.<sup>24</sup> Zeta captures the value of an asset in not only hedging the goals, but also in serving as a risky asset (that potentially earns a higher return than the goal in order to reduce the underfunding). Coqueret et al. (2017) have recently argued for examining the use of equities (other than market-cap weighted indices) in hedging bond-like liabilities, effectively capturing the dual value of equities. This explains why Zeta is a unique function of asset allocation to risky assets and the goal-hedging asset. The first term in Equation (13) is the ratio of allocations to the two risky assets for each goal given the target risk, and the second term has the hedging component.
- (viii) In short, the expected returns of an asset is impacted by all assets—whether they are substitutes or complements—much like one experiences in financial markets.
- (ix) The final asset pricing model is independent of  $\rho(T, L)$ . As Figure 1 demonstrates, the correlations are the variables that ensure these pair-wise equilibria in large part. With more goal-replicating assets, more correlation terms will be important.
- (x) Allocation recommendations are intuitive as shown in Equations (7) and (8) (and in Appendix III). Allocations to risky assets depend on the ratio of relative volatilities of the goal-replicating asset to the risky asset multiplied by a ratio of correlation terms that capture the risk tolerance of the investor (an easily identified and measured correlation term as opposed to MPT's "risk aversion" parameter). The allocation to the goal-replicating asset is solely a function of

correlation terms and the risk budget. These have been discussed in Section 1.

- (xi) GRAPM also shows why two pension funds like LACERA and NMPERA could have markedly different asset allocations, especially if their goal-replicating portfolios are different, but also because their target relative risk levels could be different. In MPT, the only difference in allocations is in the proportions allocated to M and F—not observed in practice.
- (xii) In other words, the "risk aversion" parameter of GRAPM is either the stated  $\rho(T, L)$ , as in the NMPERA or LACERA example, or the implied  $\rho(T, L)$  from measuring permitted tracking error. The volatility of the goal is also a key risk parameter and this is also a stated parameter in the IPS or observed parameter (*ex-post* volatility of the SAA). However, to gather this data for all investors across the globe is probably impossible making this a difficult model to test empirically despite its attractive features and new approach.

In summary, using the traditional absolute riskfree asset, F, and two goal-specific risk-free assets (say G1 and G2) allows one to triangulate to establish the return of any other asset. G1 and G2 individuals who are maximizing risk-adjusted returns subject to a clearly articulated risk budget will each have a unique demand for the risky asset, but the equilibrium conditions across all goals ensure that expected returns of the risky asset will stabilize at a level compatible with both goals. This equilibrium removes the need for nonobservable facets of MPT like the "risk aversion" parameter (needed for optimal asset allocation decisions) or the "market portfolio" (needed for the asset pricing model and the asset allocation decision).

In a similar fashion, having heterogeneous investors (e.g., one focused on absolute wealth, one focused on retirement, and one focused on saving for a child's college education) who are supplied their goal-specific risk-free asset and seek to maximize the relative risk-adjusted performance of their portfolios is sufficient to solve out the asset pricing model. In a heterogeneous investors model, G1 individuals are one type of investor, and G2 individuals are a different type of individual and it is clear that risky assets can be priced based on either goal.<sup>25</sup>

#### 6 Extensions/Shortcomings

In this section, we consider some extensions or shortcomings to the model because GRAPM was developed using very simple assumptions.

#### 6.1 Extensions

The simple GRAPM modeled here ignores the weights of each type of investor in the economy and hence is independent of the weight terms.<sup>26</sup> In effect, each world can be seen as a different class of investors and the merging of the worlds essentially ensures that asset prices are determined by the interaction of these different classes of investors, but weighted by the proportion of individuals in each goal. If we include this assumption, Zeta becomes a function of the weights of each market (say  $w_{G1}$  and  $w_{G2}$ ); more explicitly,  $X_{I,G1,G2}$  and  $Y_{I,G1,G2}$  in Equations (11) and (12) include  $w_{G1}$  and  $w_{G2}$ , but the allocation formulas are unchanged.<sup>27</sup>

The next simple extension is to assume that portfolio I is made up of multiple assets as opposed to it being a single risky (or alternatively, that an investor hires multiple agents). This just makes the model a bit more complicated, but because the correlations of a portfolio of assets to a goalspecific asset is nothing more than a weighted sum of each asset correlation to the goal-specific asset, the model is easily solved. Muralidhar (2001) demonstrates how this is achieved (in the context of hiring multiple agents) and further how assets that might have been considered valuable from a diversification perspective in an absolute return– risk world may be sub-optimal in a relative risk world.<sup>28</sup> If M is the portfolio of j risky assets, such that

$$r(M) = \sum_{j} w_{j} r(j), \text{ then}$$

$$\rho(M, G1) = \frac{\sum_{j} w_{j} \rho(j, G1) \sigma(j)}{\sigma(M)} \quad (17)$$

In this multi-asset (or multi-agent) setting, the allocation to each risky asset *j* that is in this portfolio =  $a * w_i$  and can be solved iteratively. This leads to potentially an interesting new research avenue as to how the optimal risky portfolio in G1 relates to the optimal risky portfolio in G2. This offers a new twist to the notion of diversification and also has some ability to explain why many pension funds globally experienced declines in funded status in the equity crises of 2000-2002 and 2008. In short, these pensions used meanvariance optimization in an absolute return-risk space to establish optimal portfolios, while ignoring that they had a goal that essentially resembled a long-duration bond asset. Hence, assets that looked attractive from a diversification perspective in MPT (e.g., equities) proved to be highly risky and even negatively correlated to the goal during crisis events. These assets might not have been included in a GRAPM portfolio for a limited relative risk budget. Coqueret et al. (2017) attempt to address this challenge. In a related vein, a more complex extension would be to consider a joint simultaneous optimization of all goals in line with Das et al. (2018) and see how that affects the choice of optimal risky portfolios. However, this approach will have to be complemented with the target relative risk for the combined multigoal optimization, which may make this quite complex.

Third, it is possible to have different relative risk budgets for different goals as the asset pricing equations are independent of  $\rho(T, L)$ , while the asset allocations are impacted by this variable. Fourth, the model could be extended to a multiperiod model as opposed to a single period model much like Merton (1973) does for CAPM.

#### 6.2 Shortcomings

The biggest shortcoming, for those academically inclined, is the fact that this approach is not based on a utility function. That choice was explained earlier and could prove an interesting avenue for future research if an appropriate utility function could be specified. More broadly, like any economic model, GRAPM can be faulted for a number of the simplifying assumptions made to arrive at the result. First, the danger of pricing assets based on one or two goal-replicating assets that may be relatively illiquid could pose severe problems as opposed to CAPM where the pricing formula depends on a highly diversified market portfolio.<sup>29</sup> In a related vein, as with all these finance models, one of the important assumptions is that we require F as a key asset with zero volatility and zero correlation. These assumptions greatly simplify the calculations even though this asset does not truly exist. There is no easy way out of this conundrum except to use the method suggested in Black (1972), but then the formulas can get more complicated.

The next big assumption was that investors maximize goal-specific risk-adjusted performance some may disagree with this starting assumption. However, the examples of LACERA and NMPERA were used to show that such an objective might not be unreasonable in a GBI setting. Moreover, even if target relative risk is not specified, permissible relative risk is extracted by way of revealed preference (by examining the historical tracking error of agents relative to the goal of the fund).

GRAPM, like MPT or RAPM, also requires forecasts of expected returns of the goal-specific relative risk-free assets and the multiple correlations and hence is subject to the same criticism as MPT/CAPM/RAPM. Housel (2015) highlights our inability to forecast these variables and the fact that they are dynamic may require more than a one-period model. GRAPM does provide additional equations to ensure that there are pair-wise equilibria, but there is no guarantee that these can be forecast or tested empirically. Furthermore, this model ignored all supply-side issues and waved them away. Interestingly, since Mcube gives demand equations for assets, even with multiple goals (and different weights on each goal), one could derive a demand-supply equilibrium if a supply curve is specified. Finally, this model assumed that free/friction-less trading between the G1 and G2 worlds will equalize the returns of assets that are risky in both worlds. There are potential challenges to this assumption that are ignored in this simplistic model.

#### 7 Conclusions

GBI is slowly becoming the norm for investors and investors are challenged in achieving their goals because finance theory has not provided an integrated model that incorporates multiple stochastic goals and agency, while still providing investors with the three facets of investing asset pricing, asset allocation, and risk-adjusted performance measures. Individuals save for a range of goals (e.g., retirement, a child's college expenses) and each has a unique set of cash flows. Researchers have argued for the creation of a new class of relative risk-free securities for each goal. Furthermore, investors delegate to agents and are keen to ensure that the agents are skillful and hence restrict the amount of relative risk they can take relative to the goal. As a result, institutional investors seek to maximize goal relative risk-adjusted returns and specify clear absolute volatility and relative risk targets.

This paper demonstrates that, with this assumption about investor behavior, two such goals/ instruments and the traditional risk-free asset, all risky assets can be priced, without requiring a traditional utility function, risk aversion parameter, or market portfolio (none of which are easily identified in real life). Risky assets are easily priced using goal-replicating assets (which are risky for another goal). The asset pricing model is derived from the simple idea that a relatively risk-free asset for one goal is a risky asset for another, and hence these two assets, plus the absolute risk-free rate allow us to triangulate to establish returns for all other assets. In short, assuming such behavior on the part of investors creates a "pair-wise equilibrium." Adding more assets just creates a lattice of "pair-wise" equilibria. This approach also lends itself easily to an asset pricing model with heterogeneous investors and provides useful advice on asset allocation and risk-adjusted performance. It also offers new avenues for research as this is a new approach that is easily extended by relaxing some of the assumptions made to arrive at this model. However, the novelty of this approach is to derive a theoretical model based on empirical observation as to how investors actually behave and how finance science is being used to improve GBI. This might be the most important contribution of this normative model and hopefully help investors achieve their goals.

#### Acknowledgments

Thanks to the late Prof. Franco Modigliani for his help and advice on developing M-cube, and to

Prof. Robert C. Merton for his collaboration on SeLFIES and constant guidance and many ideas that helped the development of this paper. Thanks to Drs. Sanjiv Das, Campbell Harvey, Kazuhiko Ohashi (BFFS and RAPM), Paolo Pasquariello, Sudhir Krishnamurthi, Amlan Roy, Sunghwan Shin (BFFS and RAPM), Robert Savickas, Dimitri Vayanos, Wai Lee, Paul Pfleiderer, and Mark Finn for their helpful comments/collaborations. Thanks also to Dr. William Sharpe for very helpful email exchanges on deriving asset pricing models from liability-based asset allocation models (Sharpe-Tint), which served as inspiration for this paper. Thanks to attendees of the GWU Finance Seminar, where a very early version of this idea was presented. Finally, thanks to an anonymous referee for helping with extensive comments. All remaining errors are mine.

## Appendix I—Intuition behind the relative asset pricing model

Assume the unique objective function as in the standard CAPM with mean–variance preferences for individual *S*.

$$U_S(\sigma, E) = -\sigma^2 + 2k_S E_s \qquad (A.1.1)$$

The investor is interested in maximizing the mean-variance objective function (A.1.1) with respect to the relative wealth at the end of period 1, where assets (A) are divided by the goal (L).

$$R_{1} - R_{0} = \frac{A_{1}}{L_{1}} - \frac{A_{0}}{L_{0}}$$
$$= \frac{A_{0}}{L_{0}} * \frac{(1 + r(A_{S}))}{(1 + r(L_{S}))} - \frac{A_{0}}{L_{0}}$$
$$= \frac{A_{0}}{L_{0}} \times \left[\frac{(1 + r(A_{S}))}{(1 + r(L_{S}))} - 1\right] \quad (A.1.2)$$

For relatively small  $r(A_S)$  and  $r(L_S)$ , Lauterbach and Reisman (2004) use the approximation that

$$(1 + r(A_S)/(1 + r(L_S)) - 1)$$
  
 $\approx r(A_S) - r(L_S)$  (A.1.3)

Since  $A_0$  and  $L_0$  are given, we can assume that you want to maximize  $\{r(A_S) - r(L_S)\}$  subject to variance of  $\{r(A_S) - r(L_S)\}$ . This then translates into the following elements of the objective function

$$E_{S,A} = E[r(A_S) - r(L_S)]$$
 (A.1.4)

$$\sigma_{S,A} = \sigma_{[r(AS) - r(LS)]} \tag{A.1.5}$$

 $k_S$  = risk aversion of investor S

Assume that the investor hedges against  $r(L_S)$  by borrowing at r(F) and simultaneously chooses the non-hedging portfolio to obtain  $r(NL_S)$ . Then, the asset portfolio return is expressed as

$$r(A_S) = r(L_S) + r(NL_S) - r(F)$$
 (A.1.6)

Proceeding much like one would in a Sharpe (1964) model, one can solve for RAPM as

$$E[r(i)] - r(F) = \frac{\operatorname{cov}[r(M') - r(L), r(i)]}{\operatorname{var}[r(M') - r(L)]} \times E[r(M') - r(L)]$$
(A.1.7)

Where M' is the relative market portfolio on the relative mean-variance frontier and excludes L and includes F.

#### Appendix II—Derivation of *M*-Cube Risk-Adjusted Performance Measure

In an agency situation, the principal worries about the relative return of the portfolio (relative mean or r(P) - r(L)) created by the agent, where the agent's portfolio is *P* relative to the benchmark *L*. In this paradigm, the relative risk of *P* to *L* can be defined as below.

Tracking Error(P)

$$= TE(P)$$

$$= \sqrt{\begin{bmatrix} \sigma^2(P) - 2 * \rho(P, L) \\ * \sigma(P) * \sigma(L) + \sigma^2(L) \end{bmatrix}}$$
(A.2.1)

Where  $\sigma$  is the volatility and  $\rho$  is the correlation parameter.

An intelligent principal interested to examine the risk-adjusted performance of P, especially if they are evaluating multiple agents with the same benchmark, must lever/delever this portfolio with both F and L to have a specific target risk (as in Modigliani and Modigliani, 1997), and a specific relative risk as well. In other words, they can create a new asset portfolio A, which blends P, F, and L to calculate the risk-adjusted performance of P. If they do this for all agents, then they can compare the return of A of all agents to establish truly risk-adjusted performance (as all other aspects like volatility and correlation have been normalized).

Assume that individuals maximize the goal relative risk-adjusted return, r(A) - r(L), subject to 3 constraints highlighted in Equations (A.2.2), (A.2.3), and (A.2.4).

$$r(A) = a \times r(P) + (1 - a - l)$$
$$\times r(F) + l \times r(L) \quad (A.2.2)$$
$$\sigma(A) = \sigma(L) \qquad (A.2.3)$$

and 
$$TE(A) = TE(Target)$$
 (A.2.4)

where *a* is the allocation to the risky portfolio, *P*, and *l* is the allocation to the goal-hedging portfolio, *L*, and the balance of the assets is invested in the traditional risk-free asset, *F*.

We do not constrain the sign of any of these parameters. Equations (A.2.2) and (A.2.3) are the same assumptions made by Modigliani and Modigliani (1997) and Graham and Harvey (1994, 1997). In an agency paradigm, Equation (A.2.4) is an additional constraint imposed because the principal is concerned about the skill of the agent and does not want to allow unconstrainted risk (Muralidhar, 2001). In other words,

SECOND QUARTER 2021

it is the level of relative risk the principal is comfortable taking, given its minimum confidence-inskill threshold for a given time period (Ambarish and Seigel, 1996). Note that this problem can be restated for every level of TE(Target) and  $\sigma(L)$ , thereby giving a continuum of optimal allocations for various risk levels to create the relative equivalent of the mean-variance frontier.

From the constraint on tracking error (A.2.4), a unique target correlation between portfolio Aand liability L,  $\rho(T, L)$ , is identified. Using the squares of equations on tracking error and Equation (A.2.4), and substituting for  $\sigma(A) = \sigma(L)$ obtains

$$\rho(T, L) = 1 - \frac{TE(target)^2}{2 \times \sigma^2(L)} \quad (A.2.5)$$

which can then be used to solve for *a*, *l*, and *l*-*a*-*l*.

$$a = \frac{\sigma(L)}{\sigma(P)} \left[ \sqrt{\frac{[1 - \rho(T, L)^2]}{[1 - \rho(P, L)^2]}} \right]$$
(A.2.6)  
$$l = \rho(T, L) - \rho(P, L)$$
$$\times \left[ \sqrt{\frac{[1 - \rho(T, L)^2]}{[1 - \rho(P, L)^2]}} \right]$$
(A.2.7)

Notice that *a* (allocation to risky asset) is a function of relative standard deviations and the realized correlation relative to the target correlation. But what is most interesting is that the allocation to the various investment options is completely independent of returns, though one could argue that the TE(Target), and hence  $\rho(T, L)$ , is a function of the expected return per unit of risk, and this is where the expected return of *A* enters the allocation formula. The analysis recommends the optimal liability hedge given a risk budget (allocation to *l*), the amount of leverage or deleverage (*1-a-l*) and the optimal allocation to the risky portfolio (allocation to *a*).

#### Appendix III—demonstrating how *M*-cube provides a demand function for various assets assuming a range of target risk

Assume  $\sigma(L) = 10.5\%$ ,  $\sigma(P) = 10.5\%$ , and  $\rho(P, L) = 0.975$ . Then a TE(Target) = 1.5%, implies that  $\rho(T, L) \approx 0.995$ . Figure A.3.1. provides a range of tracking error targets around this TE(Target) by just varying  $\rho(T, L)$ . For low levels of relative risk (i.e., funded status or relative risk aversion is high), allocation to the liability hedge is high (dashed line) and declines thereafter as the allocation to risky assets (dotted line) rises. In this setting, the non-linearity is not as obvious as in the example in Muralidhar and Shin (2013) that examines likely allocations for a corporate pension fund. The double-lined arrow provides the target correlation ( $\rho(T,L)$ ) for these TE(Target) levels. The allocation to F ranges from -1%to 0% for these settings but again are substantially different from zero in Muralidhar and Shin (2013).

#### Appendix IV—Deriving GRAPM

STEP 1: Use *M*-cube to establish optimal allocations to risky, goal-replicating, and risk-free assets

$$a_{T/L}^{P} = \frac{\sigma(L)}{\sigma(P)} \left[ \frac{\varphi(T, L)}{\varphi(P, L)} \right], \text{ where}$$
$$\varphi(I, J) = \sqrt{\left[1 - \rho(I, J)^{2}\right]} \quad (A.4.1)$$

$$l_{T/L}^{L} = \rho(T, L) - \rho(P, L) \left[ \frac{\varphi(T, L)}{\varphi(P, L)} \right]$$
(A.4.2)

**STEP 2:** Establish the risk-adjusted return of portfolio *A* in G1 world with*I*as risky asset  $(E[r(A)_{I,G1}])$ 

$$[r(A)_{I,G1}] = \left\{ \frac{\sigma(G1)}{\sigma(I)} \left[ \frac{\varphi(T,G1)}{\varphi(I,G1)} \right] \right\}$$

E



**Figure A.3.1** Allocation to risky assets, goal-replicating asset and absolute risk-free asset for different target tracking errors.

$$\times E[r(I)] + \left\{ \rho(T, G1) - \rho(I, G1) \\ \times \left[ \frac{\varphi(T, G1)}{\varphi(I, G1)} \right] \right\} \times E[r(G1)] \\ \times \left[ \frac{\varphi(T, G1)}{\varphi(I, G1)} \right] \right\} \times E[r(G1)] \\ + \left( 1 - \frac{\sigma(G1)}{\sigma(I)} \left[ \frac{\varphi(T, G1)}{\varphi(I, G1)} \right] \\ + \left( 1 - \frac{\sigma(G1)}{\sigma(I)} \left[ \frac{\varphi(T, G1)}{\varphi(I, G1)} \right] \right] \\ \times E[r(G1)] + \left( 1 - \frac{\sigma(G1)}{\sigma(G2)} \\ \times \left[ \frac{\varphi(T, G1)}{\varphi(G2, G1)} \right] \\ \times \left[ \frac{\varphi(T, G1)}{\varphi(I, G1)} \right] \right\} \right) \times r(F) \\ (A.4.3) \\ \times \left[ \frac{\varphi(T, G1)}{\varphi(G2, G1)} \right] \right\} ) \times r(F)$$
(A.4.3)
$$\times \left[ \frac{\varphi(T, G1)}{\varphi(G2, G1)} \right] \right\} ) \times r(F)$$

**STEP 3:** Establish the risk-adjusted return of portfolio *A* in G1 world with G2 as risky asset  $(E[r(A)_{G2,G1}])$ 

 $E[r(A)_{G2,G1}]$ 

$$= \left\{ \frac{\sigma(\text{G1})}{\sigma(\text{G2})} \left[ \frac{\varphi(T,\text{G1})}{\varphi(\text{G2},\text{G1})} \right] \right\}$$

**STEP 4:** Set  $E[r(A)_{I,G1}] = E[r(A)_{G2,G1}]$  to derive  $E[r(I)_{G1}]$  as function of G1, G2, and F

If 
$$a'_{G1} = \frac{\sigma(I)}{\sigma(G2)} \left[ \frac{\varphi(I, G1)}{\varphi(G2, G1)} \right]$$
 (A.4.5)

(A.4.4)

SECOND QUARTER 2021

JOURNAL OF INVESTMENT MANAGEMENT

Not for Distribution

And 
$$l'_{G1} = \frac{\sigma(I)}{\sigma(G1)} \times \rho(I, G1)$$
  
 $- \frac{\sigma(I)}{\sigma(G1)} \times \rho(G2, G1)$   
 $\times \left[\frac{\varphi(I, G1)}{\varphi(G2, G1)}\right]$  (A.4.6)

Define  $E[r(I)_{G1}]$  as the return of asset *I* in G1.

Then 
$$E[r(I)_{G1}]$$
  
=  $a'_{G1} \times E[r(G2)] + l'_{G1} \times E[r(G1)]$   
+  $(1 - a'_{G1} - l'_{G1}) \times r(F)$  (A.4.7)

Alternatively,  $E[r(I)_{G1} - r(F)]$ 

$$= a'_{G1} \times E[r(G2) - r(F)] + l'_{G1}$$
$$\times E[r(G1) - r(F)]$$
(A.4.8)

**STEP 5:** Set  $E[r(A)_{I,G2}] = E[r(A)_{G1,G2}]$  to derive  $E[r(I)_{G2}]$  as function of G1, G2, and F

If  $a_{G2}''$ 

$$= \frac{\sigma(I)}{\sigma(G1)} \left[ \frac{\varphi(I, G2)}{\varphi(G1, G2)} \right]$$
(A.4.9)

And  $l''_{G2}$ 

$$= \frac{\sigma(I)}{\sigma(G2)} \times \rho(I, G2) - \frac{\sigma(I)}{\sigma(G2)}$$
$$\times \rho(G1, G2) \times \left[\frac{\varphi(I, G2)}{\varphi(G1, G2)}\right]$$
(A.4.10)

Then  $E[r(I)_{G2}]$ 

$$= a''_{G2} \times E[r(G1)] + l''_{G2} \times E[r(G2)] + (1 - a''_{G2} - l''_{G2}) \times r(F) \quad (A.4.11)$$
  
Or alternatively,  $E[r(I)_{G2} - r(F)] = a''_{G2} \times E[r(G1) - r(F)]$ 

+ 
$$l''_{G2} \times E[r(G2) - r(F)]$$
 (A.4.12)

# **STEP 6:** Set $E[r(I)_{G1}] = E[r(I)_{G2}]$ to derive r(F) as a function of G1 and G2

If we define

 $X_{I,G1,G2}$ 

$$= \varphi(I, G1) - \rho(I, G2) \times \varphi(I, G1)$$
$$+ \rho(G1, G2) \times \varphi(I, G2) \quad (A.4.13)$$

 $Y_{I,G1,G2}$ 

$$= \varphi(I, G2) - \rho(I, G1) \times \varphi(I, G2)$$
$$+ \rho(G1, G2) \times \varphi(I, G1) \quad (A.4.14)$$

Then

 $r(F)_{G1,G2}$ 

$$= \frac{\left\{ E[r(G1) - \frac{\sigma(G1)}{\sigma(G2)} \\ \times (\frac{X_{I,G1,G2}}{Y_{I,G1,G2}}) \times E[r(G2)] \right\}}{[1 - \frac{\sigma(G1)}{\sigma(G2)} \times (\frac{X_{I,G1,G2}}{Y_{I,G1,G2}})]}$$
(A.4.15)

Also,

$$E[r(G1) - r(F)]$$

$$= \frac{\sigma(I)}{\sigma(G1)} \times \left(\frac{X_{I,G1,G2}}{Y_{I,G1,G2}}\right)$$

$$\times E[r(G2) - r(F)] \qquad (A.4.16)$$

**STEP 7: Express** E[r(I)] as function of E[r(G1)], r(F), volatilities, and correlations

$$E[r(I) - r(F)]$$

$$= Z_{I,G1/G2} \times E[r(G1) - r(F)] \quad (A.4.17)$$

where

$$Z_{I,G1/G2} = \frac{\sigma(I)}{\sigma(G1)} \left[ \left[ \frac{\varphi(I,G2)}{\varphi(G1,G2)} \right] \right]$$

JOURNAL OF INVESTMENT MANAGEMENT

$$+ \left(\frac{Y_{\mathrm{I},\mathrm{G1,G2}}}{X_{\mathrm{I},\mathrm{G1,G2}}}\right) \times \rho(I,\mathrm{G2})$$
$$- \left(\frac{Y_{\mathrm{I},\mathrm{G1,G2}}}{X_{\mathrm{I},\mathrm{G1,G2}}}\right) \times \rho(\mathrm{G1,G2})$$
$$\times \left[\frac{\varphi(I,\mathrm{G2})}{\varphi(\mathrm{G1,G2})}\right]$$
$$(A.4.18)$$

### STEP 8: Express r(I) as function of r(G2), r(F), volatilities and correlations

$$E[r(I) - r(F)]$$
  
= Z<sub>I,G2/G1</sub> × E[r(G2) - r(F)] (A.4.19)

Also,

$$Z_{I,G2/G1} = \left(\frac{Y_{I,G1,G2}}{X_{I,G1,G2}}\right) \times \left(\frac{\sigma(G2)}{\sigma(G1)}\right) \times Z_{I,G1/G2}$$
(A.4.20)

#### Notes

- <sup>1</sup> https://www.brainyquote.com/quotes/quotes/c/confucius140548.html?src=t\_goals
- <sup>2</sup> In the interests of brevity, the survey focuses on key papers that are relevant to this discussion and potentially ignores related contributions.
- <sup>3</sup> Muralidhar (2019b) goes further as to argue that the assumption that "investors maximize the expected utility of wealth" be replaced with "investors delegate to maximize relative risk-adjusted returns."
- <sup>4</sup> In some cases these are identical, which is what we will assume for this paper. In other cases, they can be different and this just makes the problem a bit more complex.
- <sup>5</sup> In chronological order for simplicity.
- <sup>6</sup> In Merton (1973), *L* hedged the stochastic opportunity set.
- <sup>7</sup> Assuming that the investor cannot contribute/save more and because of the nature of the contract or poor regulation, bears no personal risk if the funded status continues to decline. This is not an unreasonable assumption and is observed in US-defined benefit pension funds.
- <sup>8</sup> In a deterministic world, the liability should grow at the risk-free rate to prevent arbitrage opportunities.

- <sup>9</sup> Notice further that M-square can only be used for all benchmarks other than F.
- <sup>10</sup> Which is an exogenous relative risk target value specified by the principal and derived from TE(Target) once  $\sigma(L)$  is known.
- <sup>11</sup> The differential Sharpe normalizes for tracking error, but Muralidhar (2001) shows how two delegated portfolios with identical tracking errors can imply different confidence in skill, given their correlation and volatility.
- <sup>12</sup> These papers, for simplicity, assume that the date of death is known.
- <sup>13</sup> Muralidhar *et al.* (2016a, 2016b) discuss generic inflation indices to which BFFS could be linked. SeLF-IES are linked to a very specific index—per capita consumption—so that the standard-of-living of retirees is protected.
- <sup>14</sup> Interestingly, an instrument that has been proposed with a natural supplier, GDP-linked bonds, has not had much success getting launched because there is possibly no natural goal to ensure sufficient demand. See Shiller (2012).
- <sup>15</sup> There is an extensive literature on heterogeneous investors (e.g., with different risk aversions, portfolio constraints) and asset pricing that was not covered in Table 1.
- <sup>16</sup> http://www.nmpera.org/assets/uploads/downloads/RIO/ RFP/RFP-NO.-NM-INV-001-FY19-Total-Fund-Overlay-Services.pdf
- <sup>17</sup> Appendix III shows a continuum of optimal allocations to the three key assets, F, L and M', for various TE Target levels providing a demand function for the key assets.
- <sup>18</sup> I thank Prof. Campbell Harvey for this point.
- <sup>19</sup> In just the retirement world, Social Security systems, corporate and public defined benefit pension plans are underfunded and there is evidence that even individuals are not saving enough in their defined contribution plans (Mitchell, 2017). Hence, this is a very reasonable assumption.
- <sup>20</sup> See also Jorion (2003).
- <sup>21</sup> The detailed solution is provided in Muralidhar (2016a, 2016b) and summarized in Appendix IV.
- <sup>22</sup> Some may argue that  $E[r(A)_{I,G1}] = E[r(A)_{G2,G1}]$  is not guaranteed when the two portfolios have the same absolute and relative risk, even with an identical correlation to the goal. One possibility suggested by a reader is that if r(I), r(G2), r(G1) have the same variance, but are independent, then this may not be true. However, if we plug in  $\sigma(I) = \sigma(G2) = \sigma(G1)$  and if  $\rho(G2, G1) = \rho(I, G1) = \rho(G2, I) = 0$ , then it results

in r(G2) = r(G1), which violates our assumption (vi) that goal-replicating assets are unique, and G2 cannot be risky. As a result, this is more than likely an acceptable assumption. Additionally, note that  $a_{T/G1}^I \neq a_{T/G1}^{G2}$  and similarly, for the allocation to G1, because *I* and G2 are assumed to be distinct assets. Thanks to Prof. Kazuhiko Ohashi for this insight. Alternatively, if one portfolio has a higher return for the same target volatility and tracking error, the investor will never hold the second portfolio (asset), thereby driving demand to zero.

- <sup>23</sup> Thanks to Prof. Merton for stimulating this insight.
- <sup>24</sup> Thanks to Prof. Dimitri Vayanos for this suggestion.
- <sup>25</sup> This model is much simpler than models of investors with heterogeneous expectations/information as those models typically need stylized utility functions (CARA) and assumptions about complete markets.
- <sup>26</sup> Thanks to Prof. Merton for stimulating this insight.
- <sup>27</sup> If  $w_{G1}$  is the weight of G1 and  $w_{G2}$  is the weight of G2 (such that the sum is 1),  $X = w_{G1}\varphi(I, G1) - w_{G2}\rho(I, G2) \times \varphi(I, G1) + w_{G2}\rho(G1, G2) \times \varphi(I, G2)$ and  $Y = w_{G2}\varphi(I, G2) - w_{G1}\rho(I, G1) \times \varphi(I, G2) + w_{G1}\rho(G1, G2) \times \varphi(I, G1)$
- <sup>28</sup> Muralidhar (2001) goes further to demonstrate how this approach is superior to naïve maximization of differential Sharpe ratios (also known as Information ratios). This follows because the *M*-cube measure of riskadjusted performance, unlike the Sharpe ratio or the *M*square measure of Modigliani and Modigliani (1997), is the only measure that ranks portfolios consistent with a ranking based on confidence in skill.
- <sup>29</sup> Thanks to Prof. Merton for stimulating this insight.

#### References

- Abel, A. B. (1990). "Asset Prices under Habit Formation and Catching Up with the Joneses," *American Economic Review* **80**(2), 38–42.
- Ambarish, R. and Seigel, L. (1996). "Time Is the Essence," *Risk* **9**(8), 41–42.
- Amenc, N., Martellini, L., Goltz, F., and Milhau, V. (2010)."New Frontiers in Benchmarking and Liability-Driven Investing," *EDHEC-Risk Institute* (Paris, France).
- Arrow, K. J. (1964). "The Role of Securities in the Optimal Allocation of Risk-Bearing," *The Review of Economic Studies* **31**(2), 91–96.
- Aubry, J. P. and Crawford, C. (2019). "Impact of Public Sector Assumed Return on Investment Choices," Boston College Center for Retirement Research, 63.

- Aubry, J. P., Crawford, C., and Wandrei, K. (2018). "Stability in Overall Pension Plan Funding Masks a Growing Divide," Center for State and Local Government Excellence Issue Brief.
- Black, F. (1972). "Capital Market Equilibrium with Restricted Borrowing," *Journal of Business* **45**(3), 444– 454.
- Breeden, D. and Litzenberger, R. (1978). "Prices of Statecontingent Claims Implicit in Option Prices," *The Journal* of Business 51(4), 621–651.
- Brennan, M. (1993). "Agency and Asset Pricing. Working paper," University of California Los Angeles Anderson Graduate School of Management.
- Brennan, M. and Li, F. (2008). "Agency and Asset Pricing," Unpublished Working Paper (March 11), http://papers.ssrn.com/sol3/papers.cfm?abstract\_id= 1104546.
- Campbell, J. Y. and Cochrane, J. H. (1999). "By Force of Habit: A Consumption-Based Explanation of Aggregate Stock Market Behavior," *Journal of Political Economy* **107**, 205–251.
- Cochrane, J. H. (2005). *Asset Pricing* (Princeton University Press, Princeton, NJ).
- Coqueret, G., Martellini, L., and Milhas, V. (2017). "Equity Portfolios with Improved Liability Hedging Benefits," *Journal of Portfolio Management* **43**(2), 37–49.
- Cornell, B. and Roll, R. (2005). "A Delegated Agent Asset-Pricing Model," *Financial Analysts Journal* **61**(1), 57–69.
- Cumbo, J. and Wigglesworth, R. (2019). "Their House Is on Fire': The Pension Crisis Sweeping the World," *Financial Times*, https://www.ft.com/content/c95deea4-03e2-11ea-9afa-d9e2401fa7ca
- Das, S., Markowitz, H., Scheid, J., and Statman, M. (2010). "Portfolio Optimization with Mental Accounts," *Journal of Financial and Quantitative Analysis* **45**(2), 311–334.
- Das, S., Ostrov, D., Radhakrishnan, A., and Srivastav, D. (2018). "A New Approach to Goals-Based Wealth Management," *Journal of Investment Management* 16(3), 1–27.
- Debreu, G. (1959). The Theory of Value (Wiley, New York).
- Deguest, R., Martellini, L., Milhau, V., Suri, A., and Wang, H. (2015). "Introducing a Comprehensive Risk Allocation Framework for Goals-Based Wealth Management," EDHEC Working Papers, http://facultyresearch.edhec.com/research/edhec-publications/2015/ introducing - a - comprehensive - risk - allocation - frame work-for-goals-based-wealth-management-209580.kjsp

- The Economist. (2017). Analysts Struggle to Make Accurate Long-Term Market Forecasts. Buttonwood. August 31, 2017.
- Gómez, J.-P. and Zapatero, F. (2003). "Asset Pricing Implications of Benchmarking: A Two-Factor CAPM," *European Journal of Finance* **9**, 343–357.
- Graham, J. R. and Harvey, C. R. (1994). "Market Timing Ability and Volatility Implied in Investment Newsletters," *Asset Allocation Recommendations* NBER Working Paper #4890, October 1994.
- Graham, J. R. and Harvey, C. R. (1997). "Grading the Performance of Market Timing Newsletters," *Financial Analysts Journal*, **53**(6) 54–66.
- Heaton, J. and Lucas, D. (2000). "Portfolio Choice in the Presence of Background Risk," *The Economic Journal* **110**(460), 1–26
- Housel, M. (2015). "The Blind Forecaster: Who Pays These People?," *Motley Fool* http://www.fool.com/investing/ general/2015/02/25/the-blind-forecaster.aspx.
- International Monetary Fund (IMF). (2013). *Global Financial Stability Report*. Specifically, Chapter 1 Acute Risks Reduced: Actions Needed to Entrench Financial Stability (p. 29). April 2013, Washington D.C., USA.
- Jorion, P. (2003). "Portfolio Optimization with Tracking Error Constraints," *Financial Analysts Journal* **59**(5), 70–82.
- Lauterbach, B. and Riesman, H. (2002). "Preferences on Relative Return: A Potential Explanation for Some Pricing Anomalies," *Advances in Financial Economics* 7 93–103.
- Levy, H. (2011). *The Capital Asset Pricing Model in the 21st Century* (New York: Cambridge University Press).
- Lintner, J. (1965). "The Valuation of Risk Assets and the Selection of Risky Investments in Stock Portfolios and Capital Budgets," *Review of Economic Studies* **47**(1), 13–37.
- Markowitz, H. (1952). "Portfolio Selection," *Journal of Finance* 7(1), 77–91.
- Markowitz, H. (1990). "Foundations of Portfolio Theory," *Nobel Lecture*.
- Merton, R. (1973). "Intertemporal Capital Asset Pricing Model," *Econometrica* **41**(5), 867–887.
- Merton, R. (2007). "The Future of Retirement Planning." In *The Future of Life-Cycle Saving & Investing*, Z. Bodie, D. McLeavey, and L. B. Siegel (eds.). Research Foundation of the CFA Institute, 2007, Charlottesville, VA.
- Merton, R. (2014). "The Crisis in Retirement Planning," *Harvard Business Review*, July-August 2014.

http://hbr.org/2014/07/the-crisis-in-retirement-planning/ ar/1

- Merton, R. C. and Muralidhar, A. (2017a). "Time for Retirement SeLFIES," *Investments and Pensions Europe*, 3–4.
- Merton, R. C. and Muralidhar, A. (2017b). "The SeLFIES Model," *The Economist—Letters to the Editor*, 2017.
- Mitchell, D. (2017). "Building a More Robust and Inclusive US Retirement System Amid a Changing Economy," *The Aspen Institute Financial Security Program*, September 2017. https://www.aspeninstitute.org/publications/ building-robust-inclusive-us-retirement-system-amidchanging-economy/
- Modigliani, F. and Modigliani, L. (1997). "Risk-Adjusted Performance," *Journal of Portfolio Management* **23**(2), 45–54.
- Mossin J. (1966). "Equilibrium in a Capital Asset Market," *Econometrica* **34**(4), 768–783.
- Muralidhar, A. (2000). "Risk-Adjusted Performance— The Correlation Correction," *Financial Analysts Journal* **56**(5), 63–71.
- Muralidhar, A. (2001). "Optimal Risk-Adjusted Portfolios with Multiple Managers," *The Journal of Portfolio Management* 27(3), 97–104, https://doi.org/10.3905/jpm. 2001.319805.
- Muralidhar, A. (2011). A SMART Approach to Portfolio Management: An Innovative Paradigm for Managing Risk. Great Falls, VA: Royal Fern Publishing.
- Muralidhar, A. (2016a). "GBI = Gimme Better Instruments. An Innovation to Simplify Complex Investment Approaches," *Investments and Wealth Monitor*,
- 54–57.
- Muralidhar, A. (2016b). "Goals-Based Investing, The KISS Principle, and the Case for New Financial Instruments," *Unpublished Working Paper (SSRN)* https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2715053.
- Muralidhar, A. (2019a). Can (Financial) Ignorance Be Bliss? *Financial Analysts Journal* **75**(1).
- Muralidhar, A. (2019b). "The F-Utility of Wealth: Its All Relative," *The Journal of Investment Management* **17**(2), 1–11.
- Muralidhar, A. (2020). "A Novel Approach to Asset Pricing: Pair-Wise (or Janus) Equilibrium (Feb 20, 2020)," Available at SSRN: https://ssrn.com/abstract= 3631736.
- Muralidhar, A., Ohashi, K. and Shin, S. (2014a). "The Relative Asset Pricing Model: Implications for Asset Allocation, Rebalancing, and Asset Pricing," *Journal* of Financial Perspectives (https://www.gfsi.ey.com/thejournal-of-financial-perspectives.php).

- Muralidhar, A., Ohashi, K., and Shin, S. (2014b). "The Relative Asset Pricing Model: Toward a Unified Theory of Asset Pricing," *Journal of Investment Consulting* **15**(1), 51–66.
- Muralidhar, A., Ohashi, K., and Shin, S. (2016). "The Most Basic Missing Instrument in Financial Markets: The Case for Bonds for Financial Security," *Journal of Investment Consulting* **16**(2), 34–47.
- Muralidhar, A. and Shin, S. (2013). "The Relative Asset Pricing Model—Incorporating Liabilities and Delegation to Chief Investment Officers: Version 0.1," *Journal of Investment Consulting* **14**(1), 25–40.
- Perold, A. (2004). "The Capital Asset Pricing Model," Journal of Economic Perspectives 18(3), 3–24.
- Reisman, H. and Lauterbach, B. (2004). "Keeping up with the Joneses and the Home Bias," *European Financial Management* **10**(2), 225–234.
- Sharpe, W. (1964). "Capital Asset Prices: A Theory of Market Equilibrium under Conditions of Risk," *Journal of Finance* **19**(3), 425–442.
- Sharpe, W. (1994). "The Sharpe Ratio," *Journal of Portfolio Management* **21**(1), 49–58.
- Sharpe, W. and Tint., L. (1990). "Liabilities—A New Approach," *Journal of Portfolio Management* **16**(2) (winter), 5–10.

- Shiller, R. (2012). "Tackling the World Economy," *Harvard Business Review*. January–February 2012, https://hbr.org/2012/01/tackling-the-world-economy.
- Tobin, J. (1958). "Liquidity Preference as Behavior Towards Risk," *Review of Economic Studies* **25**(2), 65–86.
- Treynor, J. L. (1966). "How to Rate Management Investment Funds," *Harvard Business Review* **43**(1), 63–75.
- Waring, B. and Whitney, D. (2009). "An Asset–Liability Version of the Capital Asset Pricing Model with a Multi-period Two-Fund Theorem," *Journal of Portfolio Management* **35**(4), 111–130.

*Keywords*: Pair-wise equilibrium; GRAPM; goals-based investing; asset pricing; asset allocation; risk-adjusted performance; heterogeneous investors; GBI; CAPM; RAPM; ICAPM; three fund separation; Zeta; relative beta; goal-replicating assets; SeLFIES; BFFS; BEST; *M*-cube; *M*-square; Sharpe ratio.

JEL Classification: G1, G2, G12